Author's response to reviews

Title: Relationship between circulating tumor cells and epithelial to mesenchymal transition in early breast cancer.

Authors:

Michal Mego (misomego@gmail.com)
Zuzana Cierna (ciernaz@gmail.com)
Pavol Janega (pavol.janega@fmed.uniba.sk)
Marian Karaba (marian.karaba@nou.sk)
Gabriel Minarik (gabriel.minarik@gmail.com)
Juraj Benca (juraj.benca@nou.sk)
Tatiana Sedlackova (tatiana.sedlackova@gmail.com)
Gabriela Sieberova (gabriela.sieberova@gmail.com)
Paulina Gronesova (paulina.gronesova@gmail.com)
Denisa Manasova (d.manasova@gmail.com)
Daniel Pindak (daniel.pindak@nou.sk)
Jozef Sufliarsky (jozef.sufliarsky@nou.sk)
Ludovit Danihel (ludovit.danihel@fmed.uniba.sk)
James M Reuben (jreuben@mdanderson.org)
Jozef Mardiak (jozef.mardiak@nou.sk)

Version: 5
Date: 5 June 2015

Author's response to reviews: see over
To: Dafne Solera  
Executive Editor  
BMC Cancer

Dear Dr. Solera,

Thank you very much for the opportunity to submit our revised manuscript named “Relationship between circulating tumor cells and epithelial to mesenchymal transition in early breast cancer.” and for the valuable suggestions/comments of the reviewers. We would like to respond to their questions.

Reviewer: Nadia Dandachi

Reviewer's report:
The authors have adequately addressed all the comments raised by the reviewer and in my opinion the manuscript has greatly benefited from these revisions.

There are only some minor comments:

Page 4, line 116: add total number of patients included in this study (N=102).
Thank you very much for this point. We added total number of patients included in this study.

Page 5, line 143: Include the listed cell lines in materials and methods.
Thank you very much for this point. We included listed cell lines in materials and methods.

Page 5, line 147: Include FOXc2.
Thank you very much for this point. We included FOXC2.

Check for consistent writing. FOXC2 or Foxc2
Thank you very much for this point. We checked and corrected FOXC2 for consistency.

Reviewer: Vera Cappelletti

Reviewer's report:
I feel the Authors have addressed all the points in the revised version.

Please check the spelling and avoid the use of didn't and can't when using scientific English
Thank you very much for this point. We checked and tried to correct these errors.

We believe that your journal meets the expectation we address here.

Thank you in advance.
Kind regards,

Dr. Michal Mego, Assoc. Prof.
Department of Medical Oncology,
Translational Research Unit
Comenius University, Faculty of Medicine,
National Cancer Institute, Klenova 1,
833 10 Bratislava, Slovak Republic,
e-mail: misomego@gmail.com